Author’s response to reviews

Title: Intravenous to oral antibiotic switch therapy: a cross-sectional study in critical care units

Authors:

Juliano Gasparetto (julianogasparetto@gmail.com)
Felipe Tuon (flptuon@gmail.com)
Dayana dos Santos Oliveira (santos.dayana@hospitalcajuru.com.br)
Tiago Zequino (zequino.tiago@hospitalcajuru.com.br)
Gabriel Pipolo (ga.rammert@gmail.com)
Gabriel Ribeiro (Gabrielsenes@hotmail.com)
Paola Benincá (paolabeninca@hotmail.com)
June Westarb Cruz (cruz.june@grupomarista.com.br)
Thyago Moraes (thyagomoraes@hotmail.com)

Version: 5 Date: 05 Jul 2019

Author’s response to reviews:

Editor Comments:

1. Please include the full name of the ethics committee (and the institute to which it belongs) that approved the study and the committee’s reference number if appropriate. Additionally, please clarify whether any administrative permissions were required to access the raw data. If so please state who granted permission in the “Ethics approval and consent to participate” section.

ANSWER: We included the name of the committee and the number of approval. And we included th citation about administrative permission.
2. Please amend the statement in the "Consent for publication" section, as this refers to consent obtained to publish potentially-identifying information (of patients/participants). If your manuscript does not contain any individual person’s identifiable data or information, please state “Not applicable” under this section.

ANSWER: We amended the statement.

3. In the Availability of data and materials section please clearly state who should be contacted if someone wants to request the data.

ANSWER: We included the contact.

4. The individual contributions of all authors to the manuscript should be specified in the Authors’ Contributions section. Currently the contributions of Paola Delai Beninca are not described. Guidance and criteria for authorship can be found here:

http://www.biomedcentral.com/submissions/editorial-policies#authorship

ANSWER: Paola Delai Beninca is “PDB” in the “Author’s contributions Section”

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

ANSWER: OK.